

Campino, Susana; Marin-Menendez, Alejandro; Kemp, Alison; Cross, Nadia; Drought, Laura; Otto, Thomas D; Benavente, Ernest Diez; Ravenhall, Matt; Schwach, Frank; Girling, Gareth; Manske, Magnus; Theron, Michel; Gould, Kelda; Drury, Eleanor; Clark, Taane G; Kwiatkowski, Dominic P; Pance, Alena; Rayner, Julian C (2018) A forward genetic screen reveals a primary role for Plasmodium falciparum Reticulocyte Binding Protein Homologue 2a and 2b in determining alternative erythrocyte invasion pathways. PLoS pathogens, 14 (11). e1007436-. ISSN 1553-7366 DOI: https://doi.org/10.1371/journal.ppat.1007436

Downloaded from: http://researchonline.lshtm.ac.uk/4650509/

DOI: 10.1371/journal.ppat.1007436

Usage Guidelines

 $Please \ refer \ to \ usage \ guidelines \ at \ http://researchonline.lshtm.ac.uk/policies.html \ or \ alternatively \ contact \ researchonline@lshtm.ac.uk.$

Available under license: http://creativecommons.org/licenses/by/2.5/



G OPEN ACCESS

Citation: Campino S, Marin-Menendez A, Kemp A, Cross N, Drought L, Otto TD, et al. (2018) A forward genetic screen reveals a primary role for *Plasmodium falciparum* Reticulocyte Binding Protein Homologue 2a and 2b in determining alternative erythrocyte invasion pathways. PLoS Pathog 14(11): e1007436. https://doi.org/10.1371/ journal.ppat.1007436

Editor: Xin-zhuan Su, National Institutes of Health, UNITED STATES

Received: July 31, 2018

Accepted: October 24, 2018

Published: November 29, 2018

Copyright: © 2018 Campino et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Phenotpying data is included in the manuscript. PacBio sequencing data for GB4 and 7G8 are freely available as part of the Pf3k Project (https://www.malariagen.net/ projects/pf3k). The sequences and variant calls for the parents and progeny of both genetic crosses are also available through MalariaGen.

Funding: This work was supported by the Wellcome Trust (090851) and the Medical

RESEARCH ARTICLE

A forward genetic screen reveals a primary role for *Plasmodium falciparum* Reticulocyte Binding Protein Homologue 2a and 2b in determining alternative erythrocyte invasion pathways

Susana Campino^{1,2}*, Alejandro Marin-Menendez¹, Alison Kemp¹, Nadia Cross¹, Laura Drought¹, Thomas D. Otto^{1,3}, Ernest Diez Benavente², Matt Ravenhall², Frank Schwach¹, Gareth Girling¹, Magnus Manske¹, Michel Theron¹, Kelda Gould¹, Eleanor Drury¹, Taane G. Clark^{2,4}, Dominic P. Kwiatkowski^{1,5}, Alena Pance¹, Julian C. Rayner¹*

 Malaria Programme, Wellcome Sanger Institute, Wellcome Genome Campus, Hinxton, Cambridge, United Kingdom, 2 Faculty of Infectious and Tropical Diseases, London School of Hygiene and Tropical Medicine, London, United Kingdom, 3 Centre of Immunobiology, Institute of Infection, Immunity & Inflammation, College of Medical, Veterinary and Life Sciences, University of Glasgow, Glasgow, United Kingdom,
Faculty of Epidemiology and Population Health, London School of Hygiene and Tropical Medicine, London, United Kingdom, 5 Wellcome Trust Centre for Human Genetics, University of Oxford, Oxford, United Kingdom

* susana.campino@lshtm.ac.uk (SC); jr9@sanger.ac.uk (JCR)

Abstract

Invasion of human erythrocytes is essential for Plasmodium falciparum parasite survival and pathogenesis, and is also a complex phenotype. While some later steps in invasion appear to be invariant and essential, the earlier steps of recognition are controlled by a series of redundant, and only partially understood, receptor-ligand interactions. Reverse genetic analysis of laboratory adapted strains has identified multiple genes that when deleted can alter invasion, but how the relative contributions of each gene translate to the phenotypes of clinical isolates is far from clear. We used a forward genetic approach to identify genes responsible for variable erythrocyte invasion by phenotyping the parents and progeny of previously generated experimental genetic crosses. Linkage analysis using whole genome sequencing data revealed a single major locus was responsible for the majority of phenotypic variation in two invasion pathways. This locus contained the PfRh2a and PfRh2b genes, members of one of the major invasion ligand gene families, but not widely thought to play such a prominent role in specifying invasion phenotypes. Variation in invasion pathways was linked to significant differences in PfRh2a and PfRh2b expression between parasite lines, and their role in specifying alternative invasion was confirmed by CRISPR-Cas9-mediated genome editing. Expansion of the analysis to a large set of clinical P. falciparum isolates revealed common deletions, suggesting that variation at this locus is a major cause of invasion phenotypic variation in the endemic setting. This work has

Research Council UK (Grant no. MR/M01360X/1 to TGC and SC). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

implications for blood-stage vaccine development and will help inform the design and location of future large-scale studies of invasion in clinical isolates.

Author summary

Plasmodium parasites cause more than 200 million cases of malaria each year. All the symptoms of malaria are caused after *Plasmodium* parasites invade human red blood cells. Once inside, they grow, multiply and break open the red blood cells to release new parasites. This cycle is repeated every 48 hours, rapidly amplifying the number of parasites and causing severe anemia and other complications. *Plasmodium falciparum*, the parasite species responsible for almost all malaria deaths, can use multiple different pathways to invade human red blood cells, but the relative importance of each is not well understood. We tested the invasion pathways used by a collection of closely related parasites and compared their genome sequences to identify the genes responsible. This analysis revealed that expression differences in two neighboring genes of the Reticulocyte Binding Homologue family are responsible for most of the variation in two invasion pathways. *P. falciparum* may use variation in these genes to avoid the immune system or adapt to specific blood groups, which has important implications for vaccine development against malaria.

Introduction

Plasmodium falciparum is an obligate intracellular parasite, unable to replicate outside a host cell. During the blood stages of its complex life cycle, *P. falciparum* parasites must transition from one erythrocyte to another. The process by which *P. falciparum* merozoites recognize and invade human erythrocytes is therefore critical for parasite survival, but also represents a brief window when it is extracellular and vulnerable to the host immune system. To guarantee its replication, the parasite has evolved a series of strategies to evade the host immune response during erythrocyte invasion. These strategies include using multiple alternative pathways to recognize erythrocytes, which are thought to allow the parasite population to survive if a specific invasion route is blocked by the immune response, or to adapt to different human erythrocyte surface polymorphisms.

These alternate invasion pathways, and the receptor-ligand interactions that specify them, have been the subject of intensive research. Studies have employed *in vitro* invasion assays, inhibitory antibodies targeting specific ligands, enzyme treated or genetically deficient erythrocytes lacking specific receptors, and *P. falciparum* lines that have been genetically manipulated to delete specific ligands (reviewed in [1]). Together, these studies have suggested that alternative invasion is largely specified by members of the *P. falciparum* Erythrocyte Binding Antigen (PfEBA) and Reticulocyte binding Homologs (PfRh) multi-gene families (reviewed in [2]). Erythrocyte receptors have been identified for several PfEBAs and PfRhs, including three members of the Glycophorin family that are receptors for PfEBAs [3–5], and Complement Receptor 1 (CD35) that acts as a receptor for PfRh4 [6, 7]. However, the *P. falciparum* genes encoding these ligands can each be deleted without compromising parasite viability, implying that these receptor-ligand interactions are not absolutely required for invasion to occur [4, 8, 9]. By contrast, other receptor-ligand interactions, specifically those between PfRh5 and its receptor Basigin [10], and between AMA1 and RON2 [11], do appear to be essential. This, coupled with detailed video microscopy studies, has led to a model where the receptor-ligand

interactions that are redundant operate at a relatively early stage during invasion [12], and perhaps provide the parasite with alternative pathways by which to invade, while those that are essential operate at later stages that are absolutely required in all strains [13].

While this picture of alternate invasion pathways involving PfRh and PfEBA ligand families is well established, it is not at all clear what molecular mechanisms underpin the phenomenon. In one case, a *P. falciparum* strain can be made to switch between invasion pathways under selection pressure or when exposed to gentle shaking, and this switch is accompanied by transcriptional up-regulation of PfRh4 [14–16]. However, this switching phenomenon is far from universal among lab-adapted isolates [15] and whether it occurs in clinical isolates is not known. Invasion phenotyping studies with *P. falciparum* clinical isolates from Africa, Asia and South America show that variation in invasion pathways, as measured by sensitivity to enzyme treatment, is universal (reviewed in [17]). However, although some of these studies have linked alternative invasion pathways to variation in the sequence or expression of specific PfRhs or PfEBAs [18–23], the relative contribution of each ligand to a given pathway is not clear; it is likely that most such pathways are controlled by multiple genes.

Reverse genetics, deleting candidate genes one at time, has been used to great effect in dissecting the relative contribution of individual PfRh and PfEBA ligands (reviewed in [1]). However, this approach by definition only considers known candidate genes. In other eukaryotic systems genetic variants affecting complex phenotypes have been identified using forward genetic approaches such as quantitative trait loci (QTL) mapping. In such approaches, two strains of known phenotype are crossed, and the phenotype and genotype of the parents and progeny are compared to identify QTLs associated with a given phenotype. Due to technical challenges and ethical considerations, only a limited number of experimental genetic crosses have been carried out in P. falciparum [24-26], although the recent development of humanized mouse models may make such crosses more routine [27]. P. falciparum genetic crosses have previously been used to identify genes responsible for drug resistance [26, 28, 29] and primate host tropism [25]. We used the parents and progeny of two experimental crosses, 7G8 (Brazil) x GB4 (Ghana) and HB3 (Honduras) x Dd2 (Laos) in order to take an unbiased approach to identify genes underpinning alternative invasion pathways. Detailed phenotyping using a twocolour flow cytometry assay to quantitate alternative invasion [30] was combined with whole genome sequencing data [31] to perform QTL analysis. Despite the presumed complexity of invasion phenotypes, in the 7G8xGB4 cross a single major locus was identified as responsible for the majority of variation in two of the most variable alternative invasion pathways.

Results

Erythrocyte invasion phenotypes of parents and progeny from the 7G8xGB4 *P. falciparum* experimental genetic cross

The parental strains of the 7G8 (Brazil) x GB4 (Ghana) experimental genetic cross were phenotyped using a two-colour assay that sensitively measures parasite invasion into enzyme treated and untreated erythrocytes [30]. There was little difference in the ability of the two parental strains to invade erythrocytes treated with trypsin (TRY) at either high or low concentrations (Fig 1A), but there was a major difference in their ability to invade chymotrypsin (CHY)treated erythrocytes. CHY treatment had no detectable effect on GB4 invasion, but reduced invasion of 7G8 parasites by >60% (p-value = 0.00013). By contrast, neuraminidase (NEU) treatment appeared to affect the GB4 parental line more than 7G8 (p-value = 0.003) (Fig 1A).

To establish whether these phenotypes are genetically inherited, the same assay was used to phenotype all 27 available independent recombinant progeny clones from the 7G8 x GB4 genetic cross. The invasion phenotypes of the progeny clones were variable, with many



Fig 1. Erythrocyte invasion efficiency of progeny clones of the 7G8 x GB4 cross. (A) Invasion rates of the 7G8 (light grey bars) and GB4 (dark grey bars) parental strains into enzyme treated or untreated cells. # p-value = 0.25, *p-value ≈ 0.012 , ** p-value = 0.003, *** p-value = 0.00013. (B, C) Invasion profiles of 27 progeny clones (black bars) and parental strains (light grey: 7G8, dark grey: GB4) into NEU (B) and CHY treated (C) cells. Percentage values are relative to invasion into untreated cells. Results represent mean values from a minimum of 3 biological and 3 technical replicates. Error bars are standard error of the mean. (D) Correlation of mean invasion rates of progeny (black diamond) and parental (7G8: light gray, GB4: dark gray) into NEU and CHY treated erythrocytes.

intermediate between the two parental lines (Figs 1B, 1C and S1), implying that these enzymesensitive invasion pathways are complex multigenic traits. However, ordering clones based on the effect of CHY treatment showed evidence of discrete groups, with the phenotypes of some clones being similar to 7G8 (JE11-WC4 in Fig 1C), others similar to GB4, while the remainder had intermediate phenotypes. This suggests that at least in the case of a CHY-responsive pathway, there may be one or two genes with strong effect sizes underlying this complex phenotype. Given the inverse link between CHY and NEU invasion observed in the parental clones, we ranked the clones in ascending order based on their CHY phenotypes, and assessed the sensitivity of invasion to NEU treatment (Fig 1B). The two invasion phenotypes showed a clear negative correlation across all clones ($R^2 = -0.504$, Fig 1D). While this relationship was not absolute, of the five progeny clones least affected by CHY treatment, four were highly affected by NEU treatment (XF12-DEV in Fig 1B), indicating that the same gene(s) may underpin both phenotypes within this cross. Some of the other phenotypes were also correlated (S2 Fig), indicating that the same genes may underly multiple pathways in this cross. The parents of another genetic cross, Dd2 (Laos) and HB3 (Honduras) are known to differ significantly in their susceptibility to NEU treatment [12], so we also tested the parents and progeny of this cross using the same assay. The effect of NEU of the progeny clones followed an almost continuous distribution between the parental phenotypes (S3 Fig) and subsequent QTL analysis identified no significant loci associated with either the NEU or CHY variable invasion phenotype (S4 Fig). This null result could be the combined effect of several genes each with small effect sizes, or epigenetic regulatory mechanisms that escape detection by the sequencing and analysis methods used here, and invasion phenotypes were not pursued further in this cross.

Genetic mapping identifies a major locus on chromosome 13 that controls invasion pathways in the 7G8xGB4 cross

Illumina sequencing data from the parents and progeny of three *P. falciparum* experimental genetic crosses have been previously used to call single nucleotide polymorphisms (SNPs), insertion/deletions (indels) and copy number variation (CNVs) with high accuracy [31]. We used QTL mapping with a curated set of SNPs from this dataset to search for loci associated with all four invasion phenotypes in the 7G8xGB4 cross. A genome-wide scan identified a locus on chromosome 13 with a highly significant LOD score for both NEU and CHY treatment phenotypes (NEU LOD = 4.2, CHY LOD = 6.7; Fig 2). The same locus was identified for both phenotypes, and variation within this locus explains the majority of phenotypic variation for both CHY (68.9% of observed variation) and NEU (51.2% of observed variation) treatment phenotypes. No other statistically significant loci were identified for CHY treatment. For NEU treatment, several minor peaks were detected on chromosomes 7, 9 and 10 but none passed the whole genome significance threshold, and only the minor peak on chromosome 10 (discussed below) contained genes previously associated with invasion (Fig 2). No significant loci were identified for the remaining enzyme treatment phenotypes. There was limited variation amongst clones after High-TRY treatment, which affected our ability to identify genetic signals. By contrast there was extensive variation after Low-TRY treatment, but no QTL signal was detected.

To fine map the major locus on chromosome 13 underpinning the NEU and CHY treatment phenotypes we used Illumina sequence data to identify recombinant crossover breakpoints in the region (Fig 3) and compared SNPs and phenotype data between the different progeny clones to identify crossovers. This approach narrowed down the locus to a region of 66.57kb, located between 1.406 and 1.473 Mb on chromosome 13. Five progeny clones have the GB4 parental allele across this region, and all have identical CHY phenotypes to the parent GB4, with erythrocyte invasion completely unaffected by CHY treatment. Four of these same clones are also those most strongly affected by NEU treatment, as highlighted in Fig 1C. None of the remaining clones were completely unaffected by CHY treatment, and all carry the 7G8 allele at this locus. These results show that this 66.57kb segment contains the primary determinants of both NEU and CHY invasion phenotypes in this genetic cross.

To search for loci that could explain the remaining unassigned fraction of variation for the NEU and CHY phenotypes, a secondary genome-wide scan was conducted, controlling for the main effects on chromosome 13 identified in the primary scan. No other genomic regions reached statistical significance for the CHY treatment phenotype, suggesting that the chromosome 13 locus contains the primary gene(s) for this phenotype in these genetic backgrounds. For the NEU treatment phenotype, a region on chromosome 10 identified as a minor locus previously became statistically significant (LOD = 4.1, threshold = 3.6, 29.3% of remaining observed variation). This locus was mapped to a region between 1377067bp and 1569779bp,





Fig 2. Genome-wide scan to detect quantitative trait loci (QTL) associated with erythrocyte invasion in the 7G8xGB4 cross. Logarithm of odds (LOD) score results for (A) NEU and (B) CHY invasion phenotypes, correlating with 5,433 SNPs across the genome generated by whole genome sequencing data. The dashed line represents the significant threshold (95%) based on 1000 permutations of the data.

containing 57 genes, several very polymorphic, including the Merozoite Protein 3 (*PfMSP3*) gene family (S5 Fig), two of which (*PF3D7_1035700* and *PF3D7_1036300*) have been shown to bind erythrocytes [32, 33]. However, given the number of genes within this secondary locus, and its relatively low effect size on only one phenotype, we focused our subsequent analysis on the major locus on chromosome 13 affecting both NEU and CHY phenotypes.

Long read sequencing identifies a deletion between the *Plasmodium* falciparum Reticulocyte Binding Protein Homologue 2a and 2b genes

The locus on chromosome 13 contains fourteen annotated genes and two pseudogenes (Fig 3). Seven of these genes are expressed during late stages of the intra-erythrocyte cycle (>35 hours), including several previously implicated in erythrocyte invasion, most notably *P. falcip-arum* Merozoite Surface Protein 7 (*PfMSP7* and a cluster of *PfMSP7*-related genes, and two genes encoding invasion ligands from the *P. falciparum* Reticulocyte Binding Protein



Fig 3. Fine mapping the invasion-associated QTL on chromosome 13. Map of the chromosome 13 region linked to CHY and NEU invasion phenotypes. By comparing the chromosomal segments inherited from each parental strain in the progeny clones (solid grey segments represent GB4 genome sequence, open diagonal grey segments represent 7G8 genome sequence) with their invasion phenotypes (phenotypes in brackets are the mean invasion rates into CHY, NEU treated erythrocytes, as shown in Fig 1), the region responsible for determining these two invasion phenotypes can be narrowed down to a 66.5kb fragment containing 14 genes.

PATHOGENS

PLOS

Homologue family (*PfRh2a* and *PfRh2b*). Illumina sequencing data identified only 5 SNPs that differed in *PfMSP7* between the two parental clones, and none in any of the *PfMSP7* homologues.

Calling variation in *PfRh2a* and *PfRh2b* genes is more complex, as the two genes are positioned head-to-head and are identical over >8kb of their sequence, differing only at their 3' ends, which encodes the carboxy-terminal 400–500 amino acids of the expressed proteins [34]. As a result, it is impossible to assign short sequence reads unequivocally to one gene or the other through the majority of their length. To identify polymorphisms specific to each gene, we sequenced each gene in the GB4 and 7G8 parental strains using Sanger/capillary technology. We also used PacBio whole genome sequencing data for these two strains to obtain long fragment reads that cover both genes [35]. These approaches identified several SNPs and two deletions in the GB4 strain. One of the deletions, spanning 156bp, was present in a repeat region in the *PfRh2b* gene close to the 3' end, which marks the boundary between the region

shared by *PfRh2a/2b* and the region unique to *PfRh2b*. Indels in this region have been reported previously and do not affect the reading frame of the encoded protein [36, 37]. A novel deletion (681bp) was identified in the non-coding region between the *PfRh2a* and *PfRh2b* genes. This deletion could only be identified in PacBio data emphasizing the difficulty of calling variants in the extremely AT-biased non-coding regions of the *P. falciparum* genome. Using the PacBio data as a reference for Illumina mapping, we confirmed that this deletion is present in all five clones that have the same CHY phenotype as GB4 and which inherited the GB4 chromosomal segment, while it was absent from all clones that inherited the 7G8 chromosomal segment.

Presence or absence of an intergenic deletion correlates with differential expression of *PfRh2a* and *PfRh2b* and differential invasion phenotypes

This deletion in the PfRh2a/2b promoter region has not been previously reported and due to its position, we hypothesized that it might affect transcription of one or both genes. Quantitative RT-PCR was used to quantitate expression of *PfRh2a*, *PfRh2b*, members of the neighboring *PfMSP7* family as well as control invasion genes. Comparing gene expression between the two parental lines, 7G8 and GB4, revealed a >80-fold decrease in the expression of *PfRh2a* and *PfRh2b* in GB4 relative to 7G8 (Fig 4A). At an absolute level, there was almost no expression of PfRh2a in GB4 parasites, and expression of PfRh2b was extremely low. By contrast, there was no significant difference in expression of *PfMSP7* or two other neighboring *PfMSP7*-like genes between the two parental lines (Fig 4A). To test whether the differences represented a general shift in the expression of invasion ligands, we compared gene expression between the lines for three other major invasion ligands previously associated with alternative invasion pathways, PfEBA175, PfEBA181 and PfRh4. All showed only minor variation, with expression increased 1.6- (PfEBA175), 1.8- (PfEBA181) and 2.5- (PfRh4) fold in 7G8 relative to GB4. Extending the analysis to progeny clones from the 7G8xGB4 genetic cross showed that all five progeny that had inherited the *PfRh2a/2b* intergenic deletion found in GB4 also had very similar *PfRh2a* and *PfRh2b* expression levels to GB4 (AL12-QE5, Fig 4B), with transcription of both genes almost absent. By contrast, four other clones that lack the intergenic deletion, like 7G8, all had PfRh2a and PfRh2b expression levels comparable to the 7G8 parent (JB8-AUD, Fig 4B), much higher than GB4. There is therefore a clear and strong association between a novel indel in the *PfRh2a* and *PfRh2b* intergenic region, specific and extreme down-regulation of expression of these two genes, and use of a CHYresistant and NEU-sensitive invasion pathway.

In order to confirm the role of *PfRh2a* and *PfRh2b* expression in the invasion phenotype differences between 7G8 and GB4, we used a CRISPR/Cas9 strategy to modify these genes in the 7G8 parent parasite line. Because the deletion in the GB4 parent line affects expression of both genes, we took advantage of the identity between *PfRh2a* and *PfRh2b* to target a resistance cassette to the coding region common to both genes, in a dual disruption strategy (S6 Fig). After selection, drug resistant parasites were cloned and genotyped by PCR using a forward primer placed in the resistance cassette and reverse primers specific for the unique regions of either *PfRh2a* or *PfRh2b*. Genotyping established that all clones had integrated the resistance cassette in both genes (S7 Fig). This integration was confirmed by whole genome sequencing using Illumina technology; coverage plots showed an absence of reads aligning to the central region of each gene, corresponding to the locations targeted for deletion (S8 Fig).

Expression of both *PfRh2a* and *PfRh2b* in the edited clones was greatly decreased compared to the parental line 7G8 and matched levels detected in GB4 (Fig 5A), suggesting the dual disruption strategy recapitulated the expression effect of the intergenic deletion in the GB4





parent. Expression of PfRh2b protein was investigated in the parent strains and the edited clones by immunofluorescence, which showed clear PfRh2b labeling in 7G8, but with no detectable protein in GB4 or the edited clones (Fig 5B). To establish whether disruption of these genes impacted the pattern of erythrocyte invasion pathway usage, the two disrupted clones were phenotyped using the two-colour invasion assay. Both clones had phenotypes similar to the GB4 strain, being significantly more CHY-resistant and NEU-sensitive than their parental 7G8 line (Fig 5C). Deletion of *PfRh2a* and *PfRh2b* in 7G8 therefore phenocopies the



Fig 5. Gene editing and phenotyping of *P. falciparum* 7**G8 strain to eliminate** *PfRh2a* and *PfRh2b* expression. *PfRh2b and PfRh2a* were targeted in 7G8 using CRISPR/Cas9 technology and several clones were obtained from the resulting transfectant population. A) Expression of *PfRh2b and PfRh2a* genes for the parental lines 7G8 and GB4 and the edited 7G8 clones. B) Immunofluorescence analysis of the parental lines 7G8 and GB4 and two of the edited 7G8 clones using a specific anti-Rh2b antibody (green), nuclei are labelled with DAPI (blue). C) Invasion phenotypes of the edited 7G8 clones and the parental lines into erythrocytes untreated (UNT) or treated with neuraminidase (NEU), low trypsin (L-TRY), high trypsin (H-TRY), neuraminidase + trypsin (NEU+TRY) and chymotrypsin (CHY).

https://doi.org/10.1371/journal.ppat.1007436.g005

invasion preferences of GB4, and confirms that these genes are responsible for the major variation in alternative invasion pathway usage between these two lines. It should be noted that the strategy used here, by recapitulating the GB4 parent and decreasing expression of both *PfRh2a* and *PfRh2b*, cannot distinguish between individual effects of these two genes.

Indels in the *PfRh2b/ PfRh2a* locus are frequently found in clinical *P. falciparum* isolates

These data suggest that a previously undiscovered mechanism, a deletion within the PfRh2a/2b intergenic region, affects expression of these genes and can radically influence invasion phenotypes. Establishing whether such deletions are also present in clinical isolates is complicated by the fact that, as noted above, the large extent of identical sequence shared by both genes affects read mapping and hence indel calling. The presence of multiple strains within clinical infections, such as occurs commonly in high transmission regions, would also significantly impact mapping. We therefore first analysed 15 high quality genomes that had been generated from long-read PacBio data, which includes both laboratory and field isolates [35]. One isolate (KH01) had a deletion of the entire PfRh2b gene as well as a neighboring pseudogene PfRh6 (Fig 6 DelA), a significant change that has been previously identified in clinical isolates [36]. A second previously identified deletion was also confirmed in three PacBio sequenced samples (KE01, GA01, GN01); a 585bp deletion in the unique 3' end of PfRh2b that maintains the PfRh2b reading frame [37] (Fig 6 Del B). Deletions in the repeat regions between the shared





Fig 6. Large insertions and deletions at the *PfRh2a/PfRh2b* **locus are common and globally distributed.** (A) Map of genomic region around *PfRh2a* and *PfRh2b*. (B) Graphical delineation of the deletions detected using Pacbio sequencing data from 15 strains sequenced as part of the Pf3k project. A deletion of the entire *PfRH2b* gene along with the neighbouring pseudogene *PfRh6* (Del A), deletion within the unique region of *PfRh2b* (Del B) or repeat regions of *PfRh2b and PfRh2a* (Del C) which all maintain the translation frame, and an intergenic deletion (Del D). (C) Global prevalence of PfRh2b indels across field isolates sequenced as part of the MalariaGEN project; (left) the total number of clinical samples from each geographical region for which *PfRh2b* indels could be counted (WAF: West Africa; EAF: East Africa; CAF: Central Africa; SAS: South Asia; ESEA: East Southeast Asia; WSEA: West Southeast Asia), (middle) frequency of clinical isolates with DelA and (right) frequency of clinical isolates with DelB.

https://doi.org/10.1371/journal.ppat.1007436.g006

and unique regions of Pf*Rh2a* and *PfRh2b* (Fig 6 Del C) were also present in multiple isolates. Finally, a larger intergenic deletion (676bp), partially overlapping the region with the deletion present in GB4, was identified in the IT strain (Brazil) (Fig 6 Del D). Searching publicly available Illumina data from the Pf3k project revealed that more than 10% of clinical isolates had one of these deletions, making indels at this locus very common. Markedly different distributions were observed, with the deletion of the whole *PfRh2b* gene (Del A) present exclusively in Asia, and the deletion in the unique region of *PfRh2b* (Del B) found primarily in Africa. Accurate estimation of intergenic deletions such as Del D with Illumina data is difficult due to the repetitive nature of the sequence and the high AT content noted above, so frequency of Del D or other intergenic deletions could not be estimated.

Discussion

Differential usage of alternative invasion pathways is one of the most widely described variable phenotypes in clinical P. falciparum isolates, alongside drug resistance and cytoadherence. Previous attempts to identify the genes responsible for specifying alternate invasion have almost exclusively taken a reverse genetics approach, interrogating candidate genes one by one. These studies have been highly informative, but given the likely multigenic nature of the phenotypes, will always lead to only partial conclusions about the relative importance of different genes. We have applied forward genetics to this problem for the first time, using the parents and progeny of a P. falciparum experimental genetic cross. Combining sensitive phenotyping of invasion into enzyme treated erythrocytes with Illumina sequencing data revealed a single locus responsible for 69% of variation in invasion into chymotrypsin (CHY) treated erythrocytes, and 51% of variation in invasion into neuraminidase (NEU) treated erythrocytes. This locus contained 15 genes, including the *PfRH2a* and the *PfRh2b* genes. The phenotypes correlated with the presence of a deletion in the intergenic region between the two genes, which lie head to head on chromosome 13, and with a concomitant severe down-regulation of expression of both genes. The primary role of these genes was confirmed by CRISPR-Cas9 genome editing, where disruption of the PfRh2a and PfRh2b genes converted the 7G8 strain to a GB4-like invasion phenotype.

While forward genetics is by definition not limited to studying only previously identified candidate genes, it is striking that the unbiased forward genetics approach identified two candidate genes, *PfRh2a* and *PfRh2b*. However, while *PfRh2a* and *PfRh2b* have been associated with alternative invasion in the past, the central nature of their role in this case would not necessarily have been predicted. The strongest candidate might have been predicted to be the sialic acid dependent ligand, EBA-175, which binds to its receptor Glycophorin A in a NEU sensitive and CHY resistant manner [5] and when deleted leads to a switch to NEU-resistant invasion pathways [9], but we found no evidence for an association at this locus.

Our data does however fit with previous experimental studies that have focused on the PfRh2a/2b locus. Initially associated with invasion nearly two decades ago [34, 38], both PfRh2a and PfRh2b are extensively proteolytically processed, and bind to erythrocytes in a chymotrypsin sensitive and neuraminidase resistant manner [39, 40]. Antibodies raised against both proteins can inhibit erythrocyte invasion, and it has been extensively studied as a potential vaccine candidate [39, 41]. In this study, low PfRh2a/2b expression was associated with an increase in invasion into CHY-treated erythrocytes and a decrease in invasion into NEU-treated erythrocytes. Deleting the PfRh2b gene in both the sialic acid-dependent 3D7 strain and the sialic acid-independent W2mef strain resulted in very similar phenotypes, increasing invasion into CHY-treated erythrocytes and decreasing invasion into NEU-treated erythrocytes and methods are subjected as a potential strain resulted in very similar phenotypes, increasing invasion into CHY-treated erythrocytes and decreasing invasion into NEU-treated erythrocytes [8, 42]. Deletion of PfRh2b alone is also associated with changes in CHY and

NEU invasion phenotypes, and domain swap experiments subsequently revealed that it is specifically the transmembrane and cytoplasmic domains of PfRh2b that are functionally important in specifying variable invasion pathways [43]. Therefore, while the indel identified in our QTL analysis is associated with decreased transcription of both *PfRh2a* and *PfRh2b*, it is likely that it is specifically the decrease in *PfRh2b* expression that leads to the change in invasion phenotypes in this genetic cross.

These results also concur with previous studies of variation in PfRh2a and PfRh2b in clinical isolates. A naturally occurring 582bp deletion in the unique region of PfRh2b, first identified in Senegalese *P. falciparum* strains and then found to be common in multiple locations around the world [37], has been associated with decreased invasion into NEU-treated erythrocytes, although only among parasites isolated from blood group O donors [20]. In addition, single nucleotide variants and other smaller indels within *PfRh2b* have been associated with NEU sensitive invasion in Brazilian isolates [19, 44] and decreased *PfRh2b* expression levels have been associated with an increase in CHY resistant invasion in Tanzanian isolates [22]. There is therefore strong evidence that variation at *PfRh2b* in clinical isolates is widespread and likely to play a primary role in determining the CHY and NEU alternative invasion pathways.

P. falciparum genetic crosses offer a robust tool to identify loci that have a large effect size on a given phenotypic trait. However, the number of progeny for each cross is relatively small, which reduces the statistical power to identify loci with small effects. Controlling for variation at the major locus on chromosome 13 identified a second locus on chromosome 10 that was specifically in linkage with the NEU treatment phenotype, but not the CHY treatment phenotype. This locus contains most notably the PfMSP3 gene family [45], two members of which, MSP3DBL1 (MSP3.4) and MSP3DBL2 (MSP3.8) have been shown to bind erythrocytes [32, 33] as well as IgM [46]. These genes are amongst the most polymorphic in the *P. falciparum* genome and are under extremely strong balancing selection, which may be due to host acquired immunity or interaction with variable erythrocytes receptors [47]. While these, and other genes within the *PfMSP3* family, have been widely associated with invasion, they have not been previously implicated in specifying alternative invasion pathways, emphasizing the utility of unbiased forward genetics for exploring such multigenic traits. Confirmation of which genes within this large locus contribute to invasion into NEU treated erythrocytes will require further work, but members of the PfMSP3 gene family will be the obvious place to start.

The suggestion that alternative pathways are generally multigenic is emphasized by the fact that for several phenotypes, progeny clones were more affected by enzyme treatment than either parent (Fig 1C, S1 Fig). While this could be because there are multiple genes each of which has a minor effect, it might also indicate that epigenetic effects may be involved and would not be detected using our sequence-based approach. At least some invasion-associated genes are known to be clonally variant [48], with repression of invasion genes including PfRh4 associated with H3K9me3-based heterochromatin [49], and activation associated with H3K9ac [50]. Knockdown of a specific bromodomain protein predicted to bind H3K9ac, PfBDP1, results in up-regulation of a wide range of invasion-associated genes, including PfRhs and *PfEBAs* [51]. Epigenetics therefore clearly play a strong role in regulating invasion associated-genes, and as the progeny of a genetic cross have gone through both mosquito and liver stages after the parental lines were crossed, there has been ample opportunity for resetting of the epigenetic code. Given this, the fact that the PfRh2b/ PfRh2a effect reached genome-wide significance in the 7G8xGB4 cross is in some ways surprising, and indicative of its importance. Alternative approaches, such as characterizing the transcriptome, histone modifications and epigentic landscape, may be required to uncover other genes underpinning alternative invasion in future studies.

While large-scale systematic studies will be needed to completely disentangle the relative contribution of all genes involved, both reverse and forward genetic studies, and lab and field studies, are clearly converging on *PfRh2b* as a primary ligand controlling alternative invasion phenotypes. Variations within the *PfRh2b* gene exist globally, some strongly differentiated in frequency between geographic regions. The receptor for PfRh2b is not known, and it is therefore not clear what selection pressure is driving differentiation at the *PfRh2b* locus, but it is presumably linked to variation in the expression or sequence of an erythrocyte surface protein. Establishing the specific host-parasite interaction involved will be the critical next step in order to establish why variable erythrocyte invasion is such a common, and globally distributed, phenotype among *P. falciparum* strains, and to inform the design of strain-transcending vaccines.

Materials and methods

Ethics statement

Erythrocytes for culture of *Plasmodium falciparum* parasites were sourced from NHS Blood and Transplant, Cambridge, UK. All samples were anonymized. Use of erythrocytes from human donors for *P. falciparum* culture was approved by the NHS Cambridgeshire 4 Research Ethics Committee (REC reference 15/EE/0253) and the Wellcome Trust Sanger Institute Human Materials and Data Management Committee.

In vitro culture of P. falciparum

P. falciparum clones from the 7G8 x GB4 (n = 27) and HB3 x Dd2 (n = 35) crosses were kindly provided by Dr. Karen Hayton, Professor Thomas Wellems and Dr Mike Ferdig. Parasites were cultured in complete medium containing 10% human sera, at 5% hematocrit in O+ erythrocytes. Cultures were maintained at 37°C under an atmosphere of 1% O2, 3% CO2, and 96% N2. Before the performance of the invasion assay, parasite cultures were synchronized with 5% D-sorbitol (Sigma-Aldrich, Dorset, UK). DNA extraction for sequencing was undertaken using the QIAamp DNA Blood Midi Kit (Qiagen) or the Genomic tip (Qiagen) to obtain high-molecular-weight DNA.

Invasion assays

Erythrocyte invasion assays were performed as described previously [30]. Briefly, uninfected erythrocytes were treated with proteolytic or glycolytic enzymes to remove a subset of cell-surface receptors: neuraminidase (NEU) (final concentration of 20 mU/mL), trypsin (TRY) (50 µg/mL for low trypsin or 1 mg/mL for high trypsin), chymotrypsin (CHY) (1 mg/mL) and a combination of NEU and TRY. Treated erythrocytes were labeled with the intracellular dye DDAO-SE (Invitrogen), washed and incubated for 48 hours with an equal volume of P. falciparum ring stage parasites (c. 2% parasitemia). After co-incubation, parasites were labeled with the fluorescent DNA-intercalating dye Hoechst 33342 (Invitrogen), and parasites that had invaded labeled cells quantified by two color flow cytometry (BD LSRII- BD Biosciences). Stained samples were examined with a 355 nm UV laser (20 mW) and a 633 nm red laser (17 mW). Hoechst 33342 was excited using the UV laser and detected with a 450/50 filter and the DDAO-SE was excited using the red laser and detected with a 660/20 filter. A total of 100,000 events were collected for each sample. The data collected was analyses with FlowJo. Three technical replicates were performed for each enzyme/strain combination (i.e. three wells of the same combination performed in a given assay), and a minimum of 2 biological replicates were performed (i.e. each strain was assayed at least twice, on two different days). Data is presented as the mean \pm the standard error of the mean. Invasion efficiency is represented as the

percentage of invasion into enzyme treated labeled cells relative to the percent invasion into untreated labeled cells.

Genome sequencing

All parent and progeny samples were sequenced on the Illumina Genome Analyzer platform (www.illumina.com) at the Wellcome Sanger Institute. QTL analysis was performed using a subset of quality-controlled SNPs (86,158 SNPs) that were selected from an initial set of potential SNPs by applying a series of quality filters, as described [52], as well as a subsequent set of SNPs called specifically on these experimental genetic crosses [31]. The *PfRh2a* and *PfRh2b* genes were sequenced by Sanger / capillary technology using specific primers that amplify each gene independently [53]. Fifteen *P. falciparum* strains, including the 7G8 and GB4 parental strains were also sequenced using the PacBio sequencing system as part of the Pf3k Project (https://www.malariagen.net/projects/pf3k). Genome sequence data are available in ftp://ftp. sanger.ac.uk/pub/pathogens/Plasmodium/falciparum/PF3K/PilotReferenceGenomes/ DraftAnnotation/ and is summarized in [35]. Illumina data for 2,500 *P. falciparum isolates*, part of the Pf3k project, was screened to identify deletions in the *Pfrh2b* and *PfRH2a* genes.

QTL analysis

Genome wide linkage scans were performed using 5,433 SNPs for the 27 clones of the 7G8 x GB4 cross and 4,407 SNPs for 30 clones of the HB3 x Dd2 cross. Five clones within the latter cross were removed due to low sequence coverage. The SNP sets were all polymorphic between the parental clones, and there were at least two minor alleles in progeny clones. QTL analysis was performed using the *R/QTL* software (www.rqtl.org). An average recombination of 13.5 kb/cM has been described for the crosses used here [31], giving an estimate of at least 3 markers per uniquely inherited genome segment (excluding telomeres, centromeres and sub-telomeres). Thresholds for statistical significance across the genome scans were determined using permutation analysis. To identify any additional loci, secondary scans were performed and the models included loci identified in the first QTL.

Gene expression

Synchronized cultures enriched in schizonts were used to obtain RNA from the samples. The Ambion Ribopure blood kit and Isolate II RNA mini kit (Bioline, UK) was used to perform RNA extraction, followed by DNAse treatment and cDNA synthesis (Ambion). Gene expression for *PfRh2b*, *PfRh2a*, *PfAMA1*, *PfEBA175*, *PfEBA181* and *PfRh4* was measured using the conditions and primers/probes listed in [19] using qPCR Roche equipment and solutions. For *PfMSP7*, *PfMSRP1* and *PfMSRP2* primers and probes (labeled with 6-FAM) were: PfMSP7F: 5'-tgtcgattctcctcttg-3', PfMSP7R: 5'- gcacaaagtgaaacagatac-3', PfMSP7P: 5'-tcttgtcgttgtgatactcttgtgtgatactcttgt-3', PfMSRP1F: 5'-tctcttggttgtgattc-3', PfMSRP1R: 5'- gcacaagtgcaacagtgcaagtgcaagtac-3', PfMSRP2R: 5'- gggaatca gaagataatacaa-3', PfMSRP2P: 5'- ccaaagtccaaggtgctcaagtt-3'. Relative differences in gene expression were calculated using the $\Delta\Delta$ Cq method, with expression of the *PfAMA1* gene acting as a control to confirm that parasites were at a similar stage of development. The GB4 parental strain was used as the reference strain.

Immunofluorescence assay (IFA)

Air-dried thin films of late-stage schizonts were fixed in 4% formaldehyde. Fixed parasites were permeabilised in 1% Triton X100/PBS and blocked in 3%BSA/ 10% goat serum/PBS.

Parasites were probed with polyclonal rabbit anti-Rh2b (4D1, [34]) primary antibody at 1:200 overnight at 4°C. After three washes, the parasites were incubated with a fluorescent secondary antibody (Goat anti-Rabbit IgG (H+L) Highly Cross-Adsorbed Secondary Antibody, Alexa Fluor 488 #A11034) diluted 1:500 for 1 hr at room temperature. The samples were washed an additional three times and mounted in Prolong Gold (Molecular Probes) with DAPI. Antibody probing and wash steps used 3% BSA/PBS buffer. Images were captured on a Leica DMi8 fluorescent microscope and processed using Leica LAS X software and Photoshop.

Long read assemblies and indel frequency calculation

The genome sequences (PacBio data) for GB4, 7G8 and the 13 other strains were mapped to the reference *P. falciparum* genome (3D7 version 3.0) using *bwa-mem* [54]. The alignments were visualized in *BAMview* [55] to identify indel variants. For field isolates, mapped Illumina reads from the Pf3k samples (n = 2,500) were counted within and surrounding candidate regions (e.g. *PfRH2a* and *PfRH2b* genes) using *Samtools* software [56]. The coverage counts were normalized to 100-fold for presentation. Deletions were scored by generating ratios of coverage between regions within and flanking the potential deletion. Score cut-offs were chosen based on empirical distributions, with calibration using known deletions, and accounting for any mixed infections.

Genome editing

7G8 parasites were transfected by erythrocyte preloading [57]. Plasmids were obtained from PlasmoGEM: pCC1, containing the selection cassette, a barcode and homology arms for the common *PfRh2a/PfRh2b* region; pDC2 with two gRNAs targeting this region of the two genes and Cas9 (S6 Fig). 50µL of each plasmid were used to electroporate 300µL of packed RBCs at 0.31kV and 950µF with a Gene Pulser (BioRad, Watford UK). The transfected RBCs were mixed with a parasite culture of 5–6% parasitaemia containing approximately 1.5% schizonts. Targeted parasites were selected with 2.5nM WR99210 48 hours after transfection for 6 days and the culture was followed until the appearance of parasites. To ensure editing and eliminate plasmid episomes, the culture was treated with 2.5nM WR99210 together with 40µM 5FC for a week, after which the parasites were cloned by serial dilution. The clones obtained were confirmed by genotyping with the following primers and sequencing: CAM5 F 5'-ccaataga-taaaatttgtagag-3', AR1 R 5'-aggtttaatatcgacgagtc-3', AR2R 5'-gaacatcatcattcggttc-3', BR1 R5'-cgctttcgtaatttcactg-3', BR2R 5'-ctagcatcacgttggtc-3'.

Supporting information

S1 Fig. Erythrocyte invasion efficiency of 7G8xGB4 cross progeny clones into trypsin treated cells. Invasion profiles of 27 progeny clones (black bars) and parental strains (light grey: 7G8, dark grey: GB4) into erythrocytes treated with (A) low trypsin ($50 \mu g/mL$) or (B) high trypsin (1 mg/mL). Percentage values are relative to invasion into untreated cells. Results represent mean values from a minimum of 3 biological and 3 technical replicates. (JPG)

S2 Fig. Correlations between invasion phenotypes. Correlation of mean invasion rates of progeny (black diamond) and parental (7G8: light gray, GB4: dark gray) into enzyme treated erythrocytes. All phenotypes were compared in a pair-wise manner: NEU/L-TRY (A), CHY/L-TRY (B), NEU-H-TRY (C), CHY-H-TRY (D) and L-TRY-H-TRY (E). (JPG)

S3 Fig. Erythrocyte invasion efficiency of progeny clones of the HB3 x Dd2 cross. (A) Invasion rates of the Dd2 (light grey bars) and HB3 (dark grey bars) parental strains into enzymatic treated and untreated cells. *p-value> 0.06, ** p-value< 0.00001. (B) Invasion profiles of 35 progeny clones (black bars) and parental strains into NEU (B) and Low(L)-TRY treated (C) cells. Percentage values are relative to invasion into untreated cells. Results represent a minimum of 2 biological and 3 technical replicates. Error bars are standard error of the mean. (JPG)

S4 Fig. Genome-wide scan to detect quantitative trait loci (QTL) associated with erythrocyte invasion in the HB3xDd2 cross. Logarithm of odds (LOD) score results for (A) NEU and (B) CHY invasion phenotypes, correlating with 5,433 SNPs across the genome generated by whole genome sequencing data. The dashed line represents the significant threshold (95%) based on 1000 permutations of the data. No loci reached genome wide significance. (JPEG)

S5 Fig. Genome-wide scan to detect quantitative trait loci (QTL) associated with NEU sensitive erythrocyte invasion, controlling for the major locus on chromosome 13. Logarithm of odds (LOD) score results for the invasion phenotype into NEU-treated (A) and CHY-treated (B) cells correlated with 5,433 SNPs across the genome generated by whole genome sequencing data, after variation at the major locus on chromosome 13 was controlled for. The dashed line represents the significant threshold (95%) based on 1000 permutations of the data. Only a single locus reached genome-wide significance, on chromosome 10. (C) Expanded view of a chromosome 10 region showing the broad peak of association with NEU phenotype, which spans 57 genes including the *P. falciparum* Merozoite Protein 3 related multigene cluster. (TIF)

S6 Fig. Genome editing strategy targeting both *PfRh2a* and *PFRh2b* genes. A pCC1 vector was designed containing a resistance cassette for hdhfr under the control of the calmodulin promoter flanked by homology regions (HR) found within the sequence shared by both *PfRh2a* and *PfRh2b*. This vector was transfected into 7G8 parasites together with a pDC2 vector containing a CRISPR/Cas9 expression cassette and gRNAs targeting a shared sequence of *PfRh2b* and *PfRh2a*.

(TIFF)

S7 Fig. Confirmation by PCR of cassette integration into 7G8 derived clones. Genotyping PCR to confirm cassette integration in the targeted genes. The forward primer hybridises to the cassette while the reverse primers AR1 and AR2 are specific for *PfRh2a* and BR1 and BR2 primers specific for Rh2b (Wt = 7G8, 15B-25F = edited clones of 7G8). (TIFF)

S8 Fig. Confirmation by Illumina sequencing of target deletion in both *PfRh2b* and *PfRh2a* genes in 7G8 derived clones. Coverage plot of mapped Illumina reads in the parental strain 7G8 and Gb4 and two 7G8 derived clones (15D and 15E). A deep decrease in coverage is detected in both genes for the 7G8 derived clones and corresponds to the region target for deletions (red arrows), with 362bp. (JPG)

Acknowledgments

The authors would like to acknowledge advice and input from Mehdi Ghorbal and Ellen Bushell in the production of the genome editing constructs. The parental and progeny parasite lines were kindly provided by Dr. Karen Hayton, Professor Thomas Wellems and Dr Mike Ferdig.

Author Contributions

Conceptualization: Susana Campino, Julian C. Rayner.

- Formal analysis: Susana Campino, Thomas D. Otto, Ernest Diez Benavente, Matt Ravenhall, Magnus Manske, Taane G. Clark.
- Funding acquisition: Taane G. Clark, Dominic P. Kwiatkowski, Julian C. Rayner.
- **Investigation:** Susana Campino, Alejandro Marin-Menendez, Alison Kemp, Nadia Cross, Laura Drought, Thomas D. Otto, Michel Theron, Kelda Gould, Eleanor Drury, Alena Pance.
- Methodology: Susana Campino, Alejandro Marin-Menendez, Frank Schwach, Gareth Girling, Michel Theron, Alena Pance, Julian C. Rayner.

Resources: Dominic P. Kwiatkowski, Julian C. Rayner.

Supervision: Taane G. Clark, Dominic P. Kwiatkowski, Julian C. Rayner.

Visualization: Thomas D. Otto.

Writing - original draft: Susana Campino, Alena Pance, Julian C. Rayner.

Writing – review & editing: Susana Campino, Alejandro Marin-Menendez, Laura Drought, Alena Pance, Julian C. Rayner.

References

- Cowman AF, Tonkin CJ, Tham WH, Duraisingh MT. The Molecular Basis of Erythrocyte Invasion by Malaria Parasites. Cell host & microbe. 2017; 22(2):232–45. Epub 2017/08/12. https://doi.org/10.1016/ j.chom.2017.07.003 PMID: 28799908.
- Tham WH, Healer J, Cowman AF. Erythrocyte and reticulocyte binding-like proteins of Plasmodium falciparum. Trends in parasitology. 2012; 28(1):23–30. Epub 2011/12/20. S1471-4922(11)00180-2 [pii] https://doi.org/10.1016/j.pt.2011.10.002 PMID: 22178537.
- Mayer DC, Cofie J, Jiang L, Hartl DL, Tracy E, Kabat J, et al. Glycophorin B is the erythrocyte receptor of Plasmodium falciparum erythrocyte-binding ligand, EBL-1. Proceedings of the National Academy of Sciences of the United States of America. 2009; 106(13):5348–52. https://doi.org/10.1073/pnas. 0900878106 PMID: 19279206; PubMed Central PMCID: PMCPMC2664016.
- Maier AG, Duraisingh MT, Reeder JC, Patel SS, Kazura JW, Zimmerman PA, et al. Plasmodium falciparum erythrocyte invasion through glycophorin C and selection for Gerbich negativity in human populations. Nature medicine. 2003; 9(1):87–92. https://doi.org/10.1038/nm807 PMID: 12469115.
- Sim BK, Chitnis CE, Wasniowska K, Hadley TJ, Miller LH. Receptor and ligand domains for invasion of erythrocytes by Plasmodium falciparum. Science (New York, NY. 1994; 264(5167):1941–4. PMID: 8009226.
- Tham WH, Wilson DW, Lopaticki S, Schmidt CQ, Tetteh-Quarcoo PB, Barlow PN, et al. Complement receptor 1 is the host erythrocyte receptor for Plasmodium falciparum PfRh4 invasion ligand. Proceedings of the National Academy of Sciences of the United States of America. 2010; 107(40):17327–32. https://doi.org/10.1073/pnas.1008151107 PMID: 20855594; PubMed Central PMCID: PMCPMC2951459.
- Spadafora C, Awandare GA, Kopydlowski KM, Czege J, Moch JK, Finberg RW, et al. Complement receptor 1 is a sialic acid-independent erythrocyte receptor of Plasmodium falciparum. PLoS Pathog. 2010; 6(6):e1000968. https://doi.org/10.1371/journal.ppat.1000968 PMID: 20585558; PubMed Central PMCID: PMCPMC2887475.
- Duraisingh MT, Triglia T, Ralph SA, Rayner JC, Barnwell JW, McFadden GI, et al. Phenotypic variation of Plasmodium falciparum merozoite proteins directs receptor targeting for invasion of human erythrocytes. The EMBO journal. 2003; 22(5):1047–57. https://doi.org/10.1093/emboj/cdg096 PMID: 12606570.

- Reed MB, Caruana SR, Batchelor AH, Thompson JK, Crabb BS, Cowman AF. Targeted disruption of an erythrocyte binding antigen in Plasmodium falciparum is associated with a switch toward a sialic acid-independent pathway of invasion. Proceedings of the National Academy of Sciences of the United States of America. 2000; 97(13):7509–14. PMID: <u>10861015</u>; PubMed Central PMCID: PMCPMC16576.
- Crosnier C, Bustamante LY, Bartholdson SJ, Bei AK, Theron M, Uchikawa M, et al. Basigin is a receptor essential for erythrocyte invasion by Plasmodium falciparum. Nature. 2011; 480(7378):534–7. Epub 2011/11/15. nature10606 [pii] https://doi.org/10.1038/nature10606 PMID: 22080952; PubMed Central PMCID: PMC3245779.
- Lamarque M, Besteiro S, Papoin J, Roques M, Vulliez-Le Normand B, Morlon-Guyot J, et al. The RON2-AMA1 interaction is a critical step in moving junction-dependent invasion by apicomplexan parasites. PLoS Pathog. 2011; 7(2):e1001276. Epub 2011/02/25. https://doi.org/10.1371/journal.ppat. 1001276 PMID: 21347343; PubMed Central PMCID: PMC3037350.
- Weiss GE, Crabb BS, Gilson PR. Overlaying Molecular and Temporal Aspects of Malaria Parasite Invasion. Trends in parasitology. 2016; 32(4):284–95. <u>https://doi.org/10.1016/j.pt.2015.12.007</u> PMID: 26778295.
- Wright GJ, Rayner JC. Plasmodium falciparum erythrocyte invasion: combining function with immune evasion. PLoS Pathog. 2014; 10(3):e1003943. Epub 2014/03/22. https://doi.org/10.1371/journal.ppat. 1003943 PMID: 24651270; PubMed Central PMCID: PMC3961354.
- 14. Dolan SA, Miller LH, Wellems TE. Evidence for a switching mechanism in the invasion of erythrocytes by Plasmodium falciparum. The Journal of clinical investigation. 1990; 86(2):618–24. <u>https://doi.org/10.1172/JCI114753 PMID: 2200806</u>.
- Stubbs J, Simpson KM, Triglia T, Plouffe D, Tonkin CJ, Duraisingh MT, et al. Molecular mechanism for switching of P. falciparum invasion pathways into human erythrocytes. Science (New York, NY. 2005; 309(5739):1384–7. https://doi.org/10.1126/science.1115257 PMID: 16123303.
- Awandare GA, Nyarko PB, Aniweh Y, Ayivor-Djanie R, Stoute JA. Plasmodium falciparum strains spontaneously switch invasion phenotype in suspension culture. Scientific reports. 2018; 8(1):5782. Epub 2018/04/11. https://doi.org/10.1038/s41598-018-24218-0 PMID: <u>29636510</u>; PubMed Central PMCID: PMCPMC5893586.
- Authors WC, Ahouidi AD, Amambua-Ngwa A, Awandare GA, Bei AK, Conway DJ, et al. Malaria Vaccine Development: Focusing Field Erythrocyte Invasion Studies on Phenotypic Diversity: The West African Merozoite Invasion Network (WAMIN). Trends in parasitology. 2016; 32(4):274–83. <u>https://doi.org/ 10.1016/j.pt.2015.11.009 PMID: 26725306</u>.
- Mensah-Brown HE, Amoako N, Abugri J, Stewart LB, Agongo G, Dickson EK, et al. Analysis of Erythrocyte Invasion Mechanisms of Plasmodium falciparum Clinical Isolates Across 3 Malaria-Endemic Areas in Ghana. J Infect Dis. 2015; 212(8):1288–97. https://doi.org/10.1093/infdis/jiv207 PMID: 25838264; PubMed Central PMCID: PMCPMC4577045.
- Gomez-Escobar N, Amambua-Ngwa A, Walther M, Okebe J, Ebonyi A, Conway DJ. Erythrocyte invasion and merozoite ligand gene expression in severe and mild Plasmodium falciparum malaria. J Infect Dis. 2010; 201(3):444–52. Epub 2010/01/07. https://doi.org/10.1086/649902 PMID: 20050806.
- Lopez-Perez M, Villasis E, Machado RL, Povoa MM, Vinetz JM, Blair S, et al. Plasmodium falciparum field isolates from South America use an atypical red blood cell invasion pathway associated with invasion ligand polymorphisms. PloS one. 2012; 7(10):e47913. https://doi.org/10.1371/journal.pone. 0047913 PMID: 23118907; PubMed Central PMCID: PMCPMC3485327.
- Jennings CV, Ahouidi AD, Zilversmit M, Bei AK, Rayner J, Sarr O, et al. Molecular analysis of erythrocyte invasion in Plasmodium falciparum isolates from Senegal. Infection and immunity. 2007; 75 (7):3531–8. https://doi.org/10.1128/IAI.00122-07 PMID: 17470537.
- Bei AK, Membi CD, Rayner JC, Mubi M, Ngasala B, Sultan AA, et al. Variant merozoite protein expression is associated with erythrocyte invasion phenotypes in Plasmodium falciparum isolates from Tanzania. Molecular and biochemical parasitology. 2007; 153(1):66–71. https://doi.org/10.1016/j.molbiopara. 2007.01.007 PMID: 17303262.
- Lantos PM, Ahouidi AD, Bei AK, Jennings CV, Sarr O, Ndir O, et al. Erythrocyte invasion profiles are associated with a common invasion ligand polymorphism in Senegalese isolates of Plasmodium falciparum. Parasitology. 2009; 136(1):1–9. https://doi.org/10.1017/S0031182008005167 PMID: 19126266.
- Walliker D, Quakyi IA, Wellems TE, McCutchan TF, Szarfman A, London WT, et al. Genetic analysis of the human malaria parasite Plasmodium falciparum. Science (New York, NY. 1987; 236(4809):1661–6. PMID: 3299700.
- Hayton K, Gaur D, Liu A, Takahashi J, Henschen B, Singh S, et al. Erythrocyte binding protein PfRH5 polymorphisms determine species-specific pathways of Plasmodium falciparum invasion. Cell host & microbe. 2008; 4(1):40–51. https://doi.org/10.1016/j.chom.2008.06.001 PMID: 18621009.

- Wellems TE, Panton LJ, Gluzman IY, do Rosario VE, Gwadz RW, Walker-Jonah A, et al. Chloroquine resistance not linked to mdr-like genes in a Plasmodium falciparum cross. Nature. 1990; 345 (6272):253–5. https://doi.org/10.1038/345253a0 PMID: 1970614.
- Vaughan AM, Pinapati RS, Cheeseman IH, Camargo N, Fishbaugher M, Checkley LA, et al. Plasmodium falciparum genetic crosses in a humanized mouse model. Nature methods. 2015; 12(7):631–3. Epub 2015/ 06/02. https://doi.org/10.1038/nmeth.3432 PMID: 26030447; PubMed Central PMCID: PMCPMC4547688.
- Fidock DA, Nomura T, Talley AK, Cooper RA, Dzekunov SM, Ferdig MT, et al. Mutations in the P. falciparum digestive vacuole transmembrane protein PfCRT and evidence for their role in chloroquine resistance. Mol Cell. 2000; 6(4):861–71. PMID: 11090624; PubMed Central PMCID: PMCPMC2944663.
- Su XZ, Heatwole VM, Wertheimer SP, Guinet F, Herrfeldt JA, Peterson DS, et al. The large diverse gene family var encodes proteins involved in cytoadherence and antigenic variation of Plasmodium falciparum-infected erythrocytes. Cell. 1995; 82(1):89–100. Epub 1995/07/14. PMID: <u>7606788</u>.
- Theron M, Hesketh RL, Subramanian S, Rayner JC. An adaptable two-color flow cytometric assay to quantitate the invasion of erythrocytes by Plasmodium falciparum parasites. Cytometry A. 2010; 77 (11):1067–74. Epub 2010/09/28. https://doi.org/10.1002/cyto.a.20972 PMID: 20872885; PubMed Central PMCID: PMC3047707.
- Miles A, Iqbal Z, Vauterin P, Pearson R, Campino S, Theron M, et al. Indels, structural variation, and recombination drive genomic diversity in Plasmodium falciparum. Genome research. 2016; 26 (9):1288–99. https://doi.org/10.1101/gr.203711.115 PMID: 27531718.
- 32. Hodder AN, Czabotar PE, Uboldi AD, Clarke OB, Lin CS, Healer J, et al. Insights into Duffy binding-like domains through the crystal structure and function of the merozoite surface protein MSPDBL2 from Plasmodium falciparum. The Journal of biological chemistry. 2012; 287(39):32922–39. https://doi.org/10.1074/jbc.M112.350504 PMID: 22843685; PubMed Central PMCID: PMCPMC3463323.
- Wickramarachchi T, Cabrera AL, Sinha D, Dhawan S, Chandran T, Devi YS, et al. A novel Plasmodium falciparum erythrocyte binding protein associated with the merozoite surface, PfDBLMSP. International journal for parasitology. 2009; 39(7):763–73. PMID: 19367830.
- 34. Rayner JC, Galinski MR, Ingravallo P, Barnwell JW. Two Plasmodium falciparum genes express merozoite proteins that are related to Plasmodium vivax and Plasmodium yoelii adhesive proteins involved in host cell selection and invasion. Proceedings of the National Academy of Sciences of the United States of America. 2000; 97(17):9648–53. https://doi.org/10.1073/pnas.160469097 PMID: 10920203.
- Otto TD, Bohme U, Sanders M, Reid A, Bruske EI, Duffy CW, et al. Long read assemblies of geographically dispersed Plasmodium falciparum isolates reveal highly structured subtelomeres. Wellcome Open Res. 2018; 3:52. Epub 2018/06/05. https://doi.org/10.12688/wellcomeopenres.14571.1 PMID: 29862326; PubMed Central PMCID: PMCPMC5964635.
- Cheeseman IH, Miller B, Tan JC, Tan A, Nair S, Nkhoma SC, et al. Population Structure Shapes Copy Number Variation in Malaria Parasites. Mol Biol Evol. 2016; 33(3):603–20. https://doi.org/10.1093/ molbev/msv282 PMID: 26613787; PubMed Central PMCID: PMCPMC4760083.
- Ahouidi AD, Bei AK, Neafsey DE, Sarr O, Volkman S, Milner D, et al. Population genetic analysis of large sequence polymorphisms in Plasmodium falciparum blood-stage antigens. Infection, genetics and evolution: journal of molecular epidemiology and evolutionary genetics in infectious diseases. 2010; 10(2):200–6. https://doi.org/10.1016/j.meegid.2009.11.008 PMID: 19931645; PubMed Central PMCID: PMCPMC2829370.
- Triglia T, Thompson J, Caruana SR, Delorenzi M, Speed T, Cowman AF. Identification of proteins from Plasmodium falciparum that are homologous to reticulocyte binding proteins in Plasmodium vivax. Infection and immunity. 2001; 69(2):1084–92. Epub 2001/02/13. https://doi.org/10.1128/IAI.69.2.1084-1092.2001 PMID: 11160005; PubMed Central PMCID: PMC97989.
- Sahar T, Reddy KS, Bharadwaj M, Pandey AK, Singh S, Chitnis CE, et al. Plasmodium falciparum reticulocyte binding-like homologue protein 2 (PfRH2) is a key adhesive molecule involved in enythrocyte invasion. PloS one. 2011; 6(2):e17102. https://doi.org/10.1371/journal.pone.0017102 PMID: 213868888; PubMed Central PMCID: PMCPMC3046117.
- Triglia T, Chen L, Lopaticki S, Dekiwadia C, Riglar DT, Hodder AN, et al. Plasmodium falciparum merozoite invasion is inhibited by antibodies that target the PfRh2a and b binding domains. PLoS Pathog. 2011; 7(6):e1002075. Epub 2011/06/24. https://doi.org/10.1371/journal.ppat.1002075 PMID: 21698217; PubMed Central PMCID: PMCPMC3116812.
- Gunalan K, Gao X, Liew KJ, Preiser PR. Differences in erythrocyte receptor specificity of different parts of the Plasmodium falciparum reticulocyte binding protein homologue 2a. Infection and immunity. 2011; 79(8):3421–30. https://doi.org/10.1128/IAI.00201-11 PMID: 21628513; PubMed Central PMCID: PMCPMC3147545.
- 42. DeSimone TM, Jennings CV, Bei AK, Comeaux C, Coleman BI, Refour P, et al. Cooperativity between Plasmodium falciparum adhesive proteins for invasion into erythrocytes. Molecular microbiology. 2009;

72(3):578–89. Epub 2009/04/30. https://doi.org/10.1111/j.1365-2958.2009.06667.x PMID: 19400777; PubMed Central PMCID: PMC2891881.

- Dvorin JD, Bei AK, Coleman BI, Duraisingh MT. Functional diversification between two related Plasmodium falciparum merozoite invasion ligands is determined by changes in the cytoplasmic domain. Molecular microbiology. 2010; 75(4):990–1006. https://doi.org/10.1111/j.1365-2958.2009.07040.x PMID: 20487292; PubMed Central PMCID: PMCPMC3627358.
- Lobo CA, Rodriguez M, Struchiner CJ, Zalis MG, Lustigman S. Associations between defined polymorphic variants in the PfRH ligand family and the invasion pathways used by P. falciparum field isolates from Brazil. Molecular and biochemical parasitology. 2006; 149(2):246–51. https://doi.org/10.1016/j.molbiopara.2006.05.011 PMID: 16857278.
- Singh S, Soe S, Weisman S, Barnwell JW, Perignon JL, Druilhe P. A conserved multi-gene family induces cross-reactive antibodies effective in defense against Plasmodium falciparum. PloS one. 2009; 4(4):e5410. https://doi.org/10.1371/journal.pone.0005410 PMID: 19404387.
- Crosnier C, Iqbal Z, Knuepfer E, Maciuca S, Perrin AJ, Kamuyu G, et al. Binding of Plasmodium falciparum Merozoite Surface Proteins DBLMSP and DBLMSP2 to Human Immunoglobulin M Is Conserved among Broadly Diverged Sequence Variants. The Journal of biological chemistry. 2016; 291 (27):14285–99. https://doi.org/10.1074/jbc.M116.722074 PMID: 27226583; PubMed Central PMCID: PMCPMC4933183.
- Amambua-Ngwa A, Tetteh KK, Manske M, Gomez-Escobar N, Stewart LB, Deerhake ME, et al. Population genomic scan for candidate signatures of balancing selection to guide antigen characterization in malaria parasites. PLoS genetics. 2012; 8(11):e1002992. https://doi.org/10.1371/journal.pgen. 1002992 PMID: 23133397; PubMed Central PMCID: PMCPMC3486833.
- Cortes A, Carret C, Kaneko O, Yim Lim BY, Ivens A, Holder AA. Epigenetic silencing of Plasmodium falciparum genes linked to erythrocyte invasion. PLoS Pathog. 2007; 3(8):e107. https://doi.org/10.1371/ journal.ppat.0030107 PMID: 17676953; PubMed Central PMCID: PMCPMC1937010.
- 49. Jiang L, Lopez-Barragan MJ, Jiang H, Mu J, Gaur D, Zhao K, et al. Epigenetic control of the variable expression of a Plasmodium falciparum receptor protein for erythrocyte invasion. Proceedings of the National Academy of Sciences of the United States of America. 2010; 107(5):2224–9. https://doi.org/10.1073/pnas.0913396107 PMID: 20080673; PubMed Central PMCID: PMCPMC2836689.
- Crowley VM, Rovira-Graells N, Ribas de Pouplana L, Cortes A. Heterochromatin formation in bistable chromatin domains controls the epigenetic repression of clonally variant Plasmodium falciparum genes linked to erythrocyte invasion. Molecular microbiology. 2011; 80(2):391–406. <u>https://doi.org/10.1111/j. 1365-2958.2011.07574.x</u> PMID: 21306446.
- Josling GA, Petter M, Oehring SC, Gupta AP, Dietz O, Wilson DW, et al. A Plasmodium Falciparum Bromodomain Protein Regulates Invasion Gene Expression. Cell host & microbe. 2015; 17(6):741–51. https://doi.org/10.1016/j.chom.2015.05.009 PMID: 26067602.
- Manske M, Miotto O, Campino S, Auburn S, Almagro-Garcia J, Maslen G, et al. Analysis of Plasmodium falciparum diversity in natural infections by deep sequencing. Nature. 2012; 487(7407):375–9. Epub 2012/06/23. https://doi.org/10.1038/nature11174 PMID: 22722859; PubMed Central PMCID: PMC3738909.
- Rayner JC, Tran TM, Corredor V, Huber CS, Barnwell JW, Galinski MR. Dramatic difference in diversity between Plasmodium falciparum and Plasmodium vivax reticulocyte binding-like genes. The American journal of tropical medicine and hygiene. 2005; 72(6):666–74. PMID: 15964948.
- Li H, Durbin R. Fast and accurate short read alignment with Burrows-Wheeler transform. Bioinformatics. 2009; 25(14):1754–60. Epub 2009/05/20. https://doi.org/10.1093/bioinformatics/btp324 PMID: 19451168; PubMed Central PMCID: PMC2705234.
- 55. Carver T, Bohme U, Otto TD, Parkhill J, Berriman M. BamView: viewing mapped read alignment data in the context of the reference sequence. Bioinformatics. 2010; 26(5):676–7. https://doi.org/10.1093/ bioinformatics/btq010 PMID: 20071372; PubMed Central PMCID: PMC2828118.
- 56. Li H, Handsaker B, Wysoker A, Fennell T, Ruan J, Homer N, et al. The Sequence Alignment/Map format and SAMtools. Bioinformatics. 2009; 25(16):2078–9. Epub 2009/06/10. https://doi.org/10.1093/ bioinformatics/btp352 PMID: 19505943; PubMed Central PMCID: PMCPMC2723002.
- Deitsch K, Driskill C, Wellems T. Transformation of malaria parasites by the spontaneous uptake and expression of DNA from human erythrocytes. Nucleic acids research. 2001; 29(3):850–3. Epub 2001/ 02/13. PMID: <u>11160909</u>; PubMed Central PMCID: PMCPMC30384.